Author’s response to reviews

Title: Incidence of acute-onset endophthalmitis after separate bilateral cataract surgeries less than 5 days apart

Authors:

Ying Chen (chenying811@126.com)
Yu Zhang (cassandre_zy@163.com)
Xiaodan Li (616430854@qq.com)
Hong Yan (yhongb@fmmu.edu.cn)

Version: 2 Date: 22 Dec 2018

Author’s response to reviews:

Dear Dr. Nowomiejska

We have had our manuscript, Incidence of acute-onset endophthalmitis after separate bilateral cataract surgeries less than 5 days apart, edited by an English speaking and professional scientific editor at English language editing company in China. All the changes have been highlighted in this manuscript. If you still don’t consider the edited manuscript is fluent enough in English after review, we will have the manuscript edited by your recommended language editing services.

Best wishes,

Hong Yan